First - the serious problem of the abuse of psychotropic drugs in pregnancy was raised, Second - presents psychoactive drugs and substance classifications (including: cannabis, opium, opiates, coca, cocaine, hallucinogens, NPS) and their effects on the fetus Third - statement - substance use during pregnancy is associated with an increased risk of neurodevelopmental disorders requires further research

Dear all reviewers

Thank you for reviewing and accepting the manuscript.

Dr. Akhgari